

EO-1 Weekly Report


	Earth Observing – 1

	February 1, 2012 – February 7, 2012

	DOY
	032
	–
	038

	Mission Day
	4150
	–
	4156


EO-1 SPACECRAFT

Instruments

· Scheduled 122 science Data Collection Events (DCEs) this week 

· SGS pass at 033/11:36z received 2 of 3 scenes detailed below

· PF2 pass at 034/05:32z received 0 of 2 scenes detailed below

· SGS pass at 034/12:13 received 1 of 3 scenes detailed below

· Performed Instrument calibration
· ALI Internal Calibration I at 037/00:29z

EO-1 MISSION OPERATIONS CENTER

Real-Time

· Maintained nominal spacecraft state of health
· The Lunar Calibration planned for next week is prepared and set to be performed

· The battery differential voltage has appeared to level off in the -180mV to -210mV range

· We are setting up a final TDRS pass the day before the Calibration to monitor the battery during charge to verify that the differential voltage levels are remaining steady

· This Lunar Calibration will consist of the 4 ALI/Hyperion scans but will not contain the final Atmospheric Corrector scan. Instead, after the 4th scan EO-1 will go back to Earth acquisition to reduce power usage

· Monitoring the battery differential voltage, temperature, current, voltage and state of charge

· The battery Differential Voltage has been staying in the -180mV to -210mV range
· Working with Flight Software people for EO-1 to modify the TSM limits

· Worked through the disparities of the documentation, onboard and FSW code for the TSM levels. We have come to an agreement with the Flight Software people on where the limits are currently sitting
· Determining the new limits for the TSM
· The TRMM spacecraft has the same type of battery, 50 AH Super NiCad, and routinely sees differential voltages of +/-500mV
· Cataloguing all of the systems / spare parts for EO-1 to help the property manager and prepare for the contract switch

· TrollSat (TR2) is now certified for S-Band passes and will be putting 2-3 per week in the schedule
· In the coming weeks we will be setting up X-band tests with TR2

· Continuing to monitor SGS passes to ensure they remain stable
· SGS has an equipment problem and will be down until approximately August 2012
· SGS passes being taken on the SG-1/SG-22 hybrid antenna that was used during last equipment malfunction

Mission Planning
· Reviewing the Lunar Calibration (minus 5th scan)

· Creating TDRS supports to monitor battery levels
· Looking at the steps required in creating a South Pole image

· Remaking the SOP for mission planning activities for ASPEN and ASIST SCP
Flight Dynamics
· Working on changes that need to be made with a leap second addition

· The GPS group that handles the broadcast informed us of the leap second broadcast starting

· We monitored the GPS vs Ephemeris and confirmed the broadcast has occurred

· We switched the script that converts the GPS to accommodate the leap second broadcast
· Got in contact with the TDRS people and they have confirmed the TDRS satellites that are going through a naming change

· Changing these in our FD to ensure correct TDRS usage

· Worked with FDF for getting products from their new Product Servers

· Finalizing certificate based SFTP with new product server

· Looking at the steps required to take a South Pole image

System Administration
· Working with IONet personnel to schedule quarterly scans of EO-1 systems. The tentative dates are Feb. 15-16. 

· Completed creation of script to check and send email alerts for high disk usage. This script was delivered to all operational Linux systems and was configured to run via cron.

· Corrected permission problems on both Flight Dynamics computers. The permissions for a data folder within the FreeFlyer program were edited so the FOT could implement changes related to the upcoming leap second.

· Completed the periodic reboot of the backup string of operational computers. A chkdsk/fsck is performed on each system during this routine maintenance (7 systems completed).

· Corrected the printer configuration on the EO-1 mission firewall. Removed an entry for a printer that is no longer used operationally.

· Creating a thorough property checklist to properly document all NASA and MOMS equipment. This spreadsheet also accounts for the recent MOC relocation and retired legacy systems. All MOC systems have been documented so far.

GROUND AND SPACE NETWORK

Station Downtimes 

· SGS has had an equipment malfunction and latest indication is that it will be back up next August
· SG1/22 has a communication line problem, workaround is in place, fix timing unknown
· HGS is down, unknown time to be back up

Operational Discrepancies

· SGS pass on 033/11:36z had 1 scene lost due to the HRD FEP computer losing lock on X-band CDS #260441. This is a known problem for EO-1/TERRA and we are working on a fix.
· 2 scenes complete, 1 scene lost
· PF2 pass on 034/05: 23z had 2 scenes lost due to loss of commercial power at PF2 CDS#260443
· 2 scenes lost

· SGS pass on 034/12:13 had 2 scenes lost due to the HRD FEP computer losing lock on X-band CDS#260446. This is a known problem for EO-1/TERRA and we are working on a fix.
· 1 scene complete, 2 scenes lost

Number of Real-Time Supports

[image: image1.emf]Date DOY S/X-Band S-Band Only Non-DSMC TDRSS Daily Totals

1-Feb-12 32

7 3 1 1

12

2-Feb-12 33

5 5 1 0

11

3-Feb-12 34

6 3 0 0

9

4-Feb-12 35

5 3 1 0

9

5-Feb-12 36

5 4 1 0

10

6-Feb-12 37

5 3 1 0

9

7-Feb-12 38

6 3 1 0

10

39 24 6 1 70 Weekly Totals


UPCOMING EVENTS

· Lunar Calibration with only 4 scans to save battery power

· TSM update onboard spacecraft
· TR2 X-Band pass testing
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		Date		DOY		S/X-Band		S-Band Only		Non-DSMC		TDRSS		Daily Totals

		1-Feb-12		32		7		3		1		1		12

		2-Feb-12		33		5		5		1		0		11

		3-Feb-12		34		6		3		0		0		9

		4-Feb-12		35		5		3		1		0		9

		5-Feb-12		36		5		4		1		0		10

		6-Feb-12		37		5		3		1		0		9

		7-Feb-12		38		6		3		1		0		10

		Weekly Totals				39		24		6		1		70
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